Author's response to reviews

Title: Rethinking Patient-hood: Workload, Capacity and Burden of Treatment

Authors:

Carl R May (c.r.may@soton.ac.uk)
David T Eton (eton.david@mayo.edu)
Kasey Boehmer (boehler.kasey@mayo.edu)
Katie Gallacher (katie.gallacher@glasgow.ac.uk)
Katherine J Hunt (k.j.hunt@soton.ac.uk)
Sara MacDONald (sara.macdonald@glasgow.ac.uk)
Frances S Mair (frances.mair@glasgow.ac.uk)
Christine M May (chrismay@soton.ac.uk)
Victor M Montori (montori.victor@mayo.edu)
Alison Richardson (alison.richardson@soton.ac.uk)
Anne E Rogers (a.e.rogers@soton.ac.uk)
Nathan Shippee (nshippee@umn.edu)

Version: 2

Date: 28 January 2014

Author's response to reviews:

We have made no revisions to this paper. It has been transferred from BMC medicine.